# Factors that influence variation in clinical decision-making about thrombolysis in the treatment of acute ischaemic stroke: results of a discrete choice experiment

Richard G Thomson,<sup>1</sup>\* Aoife De Brún,<sup>1</sup> Darren Flynn,<sup>1</sup> Laura Ternent,<sup>1</sup> Christopher I Price,<sup>2,3</sup> Helen Rodgers,<sup>2,3</sup> Gary A Ford,<sup>4</sup> Matthew Rudd,<sup>2,3</sup> Emily Lancsar,<sup>5</sup> Stephen Simpson<sup>6</sup> and John Teah<sup>6</sup>

 <sup>1</sup>Institute of Health and Society, Newcastle University, Newcastle upon Tyne, UK
<sup>2</sup>Stroke Unit, Wansbeck General Hospital, Northumbria Healthcare NHS Foundation Trust, North Shields, UK
<sup>3</sup>Institute of Neuroscience (Stroke Research Group), Newcastle University, Newcastle upon Tyne, UK
<sup>4</sup>Oxford University Hospitals NHS Trust, Oxford, UK
<sup>5</sup>Centre for Health Economics, Monash University, Melbourne, VIC, Australia
<sup>6</sup>The Stroke Association, Gateshead, UK

\*Corresponding author

**Declared competing interests of authors:** Darren Flynn, Gary A Ford, Helen Rodgers and Richard G Thomson have been involved in marketing activity for COMPASS, a decision aid to support thrombolysis decision-making and risk communication, which may be made commercially available, including covering the costs of technical maintenance and updating of the information content. Helen Rodgers is president-elect of the British Association of Stroke Physicians and a member of the Intercollegiate Stroke Working Party. Gary A Ford's previous institution has received research grants from Boehringer Ingelheim (manufacturer of Alteplase), and honoraria from Lundbeck for stroke-related activities. Gary A Ford has also received personal remuneration for educational and advisory work from Boehringer Ingelheim and Lundbeck; in addition, he is supported by a National Institute for Health Research Senior Investigator award.

Published January 2017 DOI: 10.3310/hsdr05040

# **Plain English summary**

## Clinical decision-making about thrombolysis

Health Services and Delivery Research 2017; Vol. 5: No. 4 DOI: 10.3310/hsdr05040

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

ntravenous thrombolysis (blood thinner or 'clotbuster') is an effective emergency treatment for patients with a stroke due to a clot which blocks a blood vessel in the brain. Although this is recommended for carefully selected patients, treatment rates vary across the UK. Since the introduction of '24/7' stroke services (services available 24 hours per day, 7 days per week), continuing differences may represent differences in clinicians' decision-making on patients' suitability for treatment. Decision-making is complex, owing to the emergency context of the decision (treatment must be given within 4.5 hours of symptom onset) and the many factors that influence the levels of benefit/risk for patients. These differences in decision-making could result in undertreatment or in the inappropriate treatment of patients who do not meet treatment criteria.

We conducted a study describing hypothetical patients with different characteristics [e.g. age, gender, seriousness of symptoms using the National Institutes of Health Stroke Scale (NIHSS)] to understand which characteristics influence doctors' decision-making. We found that patients with the following characteristics were more likely to be offered thrombolysis (compared with reference levels in brackets): stroke onset time of 2 hours 30 minutes (50 minutes); moderate pre-stroke dependency on others (moderate to severely dependent); systolic blood pressure of 185 mmHg (140 mmHg); higher stroke severity scores (NIHSS 2 without aphasia); age 85 years (65 years); and Afro-Caribbean (white). Factors linked with the decision not to offer treatment were age 95 years; symptom onset time of 4 hours 15 minutes; severe dementia (compared with no memory problems); and systolic blood pressure of 200 mmHg. Three clinician-related characteristics were linked with an increased likelihood of offering treatment (perceived strength of the evidence for the treatment; treating more patients in the past 12 months; and high discomfort with uncertainty) and one characteristic was linked to a decreased likelihood of offering treatment (comfort treating patients outside treatment criteria).

# **Health Services and Delivery Research**

ISSN 2050-4349 (Print)

ISSN 2050-4357 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: nihredit@southampton.ac.uk

The full HS&DR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Services and Delivery Research journal

Reports are published in *Health Services and Delivery Research* (HS&DR) if (1) they have resulted from work for the HS&DR programme or programmes which preceded the HS&DR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **HS&DR** programme

The Health Services and Delivery Research (HS&DR) programme, part of the National Institute for Health Research (NIHR), was established to fund a broad range of research. It combines the strengths and contributions of two previous NIHR research programmes: the Health Services Research (HSR) programme and the Service Delivery and Organisation (SDO) programme, which were merged in January 2012.

The HS&DR programme aims to produce rigorous and relevant evidence on the quality, access and organisation of health services including costs and outcomes, as well as research on implementation. The programme will enhance the strategic focus on research that matters to the NHS and is keen to support ambitious evaluative research to improve health services.

For more information about the HS&DR programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hsdr

#### This report

The research reported in this issue of the journal was funded by the HS&DR programme or one of its preceding programmes as project number 12/5001/45. The contractual start date was in June 2013. The final report began editorial review in June 2015 and was accepted for publication in November 2015. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HS&DR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health.

© Queen's Printer and Controller of HMSO 2017. This work was produced by Thomson *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

## Health Services and Delivery Research Editor-in-Chief

Professor Jo Rycroft-Malone Professor of Health Services and Implementation Research, Bangor University, UK

### **NIHR Journals Library Editor-in-Chief**

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the EME Programme, UK

## **NIHR Journals Library Editors**

**Professor Ken Stein** Chair of HTA Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andree Le May Chair of NIHR Journals Library Editorial Group (EME, HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

**Professor Matthias Beck** Chair in Public Sector Management and Subject Leader (Management Group), Queen's University Management School, Queen's University Belfast, UK

**Professor Aileen Clarke** Professor of Public Health and Health Services Research, Warwick Medical School, University of Warwick, UK

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

**Professor Geoffrey Meads** Professor of Health Sciences Research, Health and Wellbeing Research Group, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

**Professor Martin Underwood** Director, Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of members of the NIHR Journals Library Board: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: nihredit@southampton.ac.uk